Author's response to reviews

Title: "A worm in the heart": the first three-dimensional visualization of a thrombus in transit entrapped between the leads of a permanent dual-chamber pacemaker: a case report

Authors:

- Petra Maagh (Petra.Maagh@rub.de)
- Thomas Butz (Thomas.Butz@marienhospital-herne.de)
- Andreas Ziegler (andreas.ziegler@bad-krozingen.de)
- Axel Meissner (axel.meissner@marienhospital-herne.de)
- Magnus W Prull (Magnus.Prull@marienhospital-herne.de)
- Hans-J Trappe (hans-joachim.trappe@rub.de)

Version: 7 Date: 24 August 2010

Author's response to reviews:

see cover letter above